Author's response to reviews

Title: The use of implantable cardioverter defibrillators in Iceland: a retrospective population based study

Authors:

Margret Leosdottir (Margret.Leosdottir@med.lu.se)
Gudrun Reimarsdottir (gudreim@landspitali.is)
Gizur Gottskalksson (gizurgot@landspitali.is)
Bjarni Torfason (bjarnito@landspitali.is)
Margret Vigfusdottir (margvigf@landspitali.is)
David O Arnar (davidar@landspitali.is)

Version: 3 Date: 5 May 2006

Author's response to reviews: see over
Covering letter


Manuscript number: 1736633650920421

Dear Senior Assistant Editor,

The authors have made final changes to the manuscript according to requests listed in your e-mail. The changes made are listed below:

1. The manuscript has been proofread to ensure conformation to the formatting check list. No changes were needed.
2. According to suggestions, changes have been made to the Background section in the Abstract.
3. In the Abstract, Conclusion has been changed to Conclusions.
4. Abstract details have subsequently been updated on the submission page.
5. Keywords have been omitted.
6. Figure titles (Figure 1, Figure 2...) in the figure files have been omitted.
7. Figures have been cropped as closely as possible.

Thank you for accepting our manuscript in your journal!

Best wishes,

Margrét Leósdóttir
First author